Author’s response to reviews

Title: Safety and efficacy of a standardized intracameral combination of mydriatics and anesthetic for cataract surgery in type-2 diabetic patients.

Authors:

Marc Labetoulle (marc.labetoulle@aphp.fr)
Anders Behndig (anders.behndig@umu.se)
Marie-José Tassignon MD (marie-jose.tassignon@uza.be)
Rudy Nuijts (rudy.nuijts@mumc.nl)
Rita Mencucci (rita.mencucci@unifi.it)
José Luis Güell (guell@imo.es)
Uwe Pleyer (uwe.pleyer@charite.de)
Jacek Szaflik (admin@szaflik.pl)
Paul Rosen (phrosen@rocketmail.com)
Alain Bérard (alain.berard@wanadoo.fr)
Frédéric Chiambaretta (fchiambaretta@chu-clermontferrand.fr)
Béatrice Cochener-Lamard (beatrice.cochener@ophtalmologie-chu29.fr)

Version: 3 Date: 12 Feb 2020

Author’s response to reviews:

Jashan Abraham
Editors, BMC Ophthalmology

Dear Dr. Abraham,

Thank you for providing the editorial comments. We have addressed all the revisions suggested the Referees. Please find below our point-by-point responses to the comments.

We thank you and the Referees for helping refine our submission. Please do not hesitate to contact us with any further queries.

Sincerely,

Marc Labetoulle MD, PhD. Service d'Ophtalmologie, Hôpital Bicêtre, APHP, Université Paris Sud. 94275 Le Kremlin-Bicêtre, France. E-mail: marc.labetoulle@bct.aphp.fr ; Phone : 33 1 45 21 36 90

Technical Comments:

Editor Comments:

1. Trial Registration date

Please include the trial registration date in the ‘Trial Registration’ sub-section of the abstract.

Response: The trial date has been included in the subsection.

2. Availability of data and materials

In the Availability of data and materials section please clearly state who should be contacted if someone wants to request the data.

Response: The contact person for data availability has been included.

3. Authors contribution
We note that two authors have the same initials [AB]. To distinguish between them in the Authors’ Contributions section please designate them as [AB]1 and [AB]2, with [AB]1 corresponding to [Anders Behndig], chosen because this name is furthest up the author list.

Alongside this, please clarify whether you would like to use the initials BC or BCL for Béatrice Cochener-Lamard, and use the same initials throughout the authors’ contributions section for clarity.

We also note that the following authors initials have not been included in the author’s contribution section [UP] and the initials [UW] have been included but do not correspond to an author, please clarify this.

Response: All the author initials described above have been corrected. The omission for UP and the lack of inclusion of all the roles for UW as an author were both mistakes on our part and have been corrected.

4. Funding body role

In the section 'Funding', please also describe the role of the funding body (in the phase 3 study) in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response: The role of the funding body has been included.

5. Additional files

Please remove the “response to reviewers” supplementary file as it is not required at this stage.

Response: It has been removed.
6. Clean copy

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions.

Response: Clean copy provided.